Author's response to reviews

Title: Prevalence and socio-demographic correlates of stunting and thinness among Pakistani primary school children

Authors:

Muhammad Umair Mushtaq (mushtaqmu@gmail.com)
Sibgha Gull (sibgha_gul@ymail.com)
Usman Khurshid (usmankhurshid_aimc@yahoo.com)
Ubeera Shahid (mushtaqmu@gmail.com)
Mushtaq Ahmad Shad (mushtaq_shad@yahoo.com)
Arif Mahmood Siddiqui (rdsidiqi@gmail.com)

Version: 6 Date: 8 October 2011

Author's response to reviews:

We are sorry for changing the files. There were some minor typographical mistakes, and the manuscript is ready for publication now. It does not need to be seen for formatting.